Author’s response to reviews

Title: Clinical outcomes of over-the-scope-clip system for the treatment of acute upper non-variceal gastrointestinal bleeding: a systematic review and meta-analysis

Authors:

Chunyu Zhong (447166418@qq.com)
Shali Tan (957242961@qq.com)
Yutang Ren (pumcren@126.com)
MuHan Lü (52721759@qq.com)
Yan Peng (1806857826@qq.com)
Xiangsheng Fu (drfuxs@gmail.com)
Xiaowei Tang (solitude5834@hotmail.com)

Version: 3 Date: 10 Dec 2019

Author’s response to reviews:

Dear Editors,

Thank you very much for considering publication of our manuscript. After discussion with all the co-authors, we revised this manuscript seriously and carefully according to your opinions. The answers to each comment are as follows:

Editor Comments:

1. Authors’ contributions - approval
Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript in its current state. Currently, not all authors are listed as having approved the final version of the manuscript.
Answer to question 1: Thanks for this comment. We have already added sentences “All authors, ZCY, TSL, RYT, LMH, PY, FXS and TXW, have read and approved the manuscript in its current state” in the Authors' contributions section.

2. Role of Funders
Please note the role(s) of the funding body (bodies) in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript, which should be declared in the Funding section.
Answer to question 1: Thanks for this comment. We have already added sentences “The funding bodies have no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript” in the Funding of Declarations section.

3. Formatting
In order to be in line with the submission guidelines for BMC Gastroenterology research articles (found in detail here: https://bmcgastroenterol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article), please make the following formatting changes:
- Please rename the section titled Materials and Methods to Methods
- Please add in a section title "Declarations" above the Declarations section
Answer to question 3: Thanks for this comment. We have already completed the above formatting changes.

4. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Answer to question 4: Thanks for this comment. We have already uploaded our manuscript as a single, final, clean version that did not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional file also were clean versions. Figures (and additional files) remained uploaded as separate files.

In addition to correct this manuscript according to the reviewers’ advice, we also checked the manuscript time and time again to avoid grammatical or spelling errors throughout the article. Hope our corrected manuscript can be published in your journal soon. That will be our great honor.

Thank you very much! Best wishes!

yours sincerely,

Xiaowei Tang